Author's response to reviews

Title: Socio-demographic determinants and access equality to prenatal care in Italy

Authors:

  Manuela Chiavarini (chiavarini@inwind.it)
  Donatella Lanari (dlanari@stat.unipg.it)
  Liliana Minelli (lminelli@unipg.it)
  Luca Salmasi (luca.salmasi@progetti.unipg.it)

Version: 2 Date: 14 February 2013

Author's response to reviews: see over
Dear Editor,

I am submitting a revised version of the paper “Socio-demographic determinants and access equality to prenatal care in Italy, by Manuela Chiavarini, Donatella Lanari, Liliana Minelli and myself, to be considered for publication in the *BMC Health Services Research*.

With respect to the previous version we added,

1. as requested a section about competing interests and
2. a section about authors' contributions.

As suggested we would like our paper to be transferred to BMC Health Services Research.

Best wishes,

Dr. Luca Salmasi